Author’s response to reviews

Title: Examining database persistence of ISO/EN 13606 standardized Electronic Health Record extracts: relational vs. noSQL approaches

Authors:
Ricardo Sánchez-de-Madariaga (ricardo.sanchez@isciii.es)
Adolfo Muñoz (adolfo.munoz@isciii.es)
Raimundo Lozano-Rubí (rlozano@clinic.ub.es)
Pablo Serrano-Balazote (pserranob@salud.madrid.org)
Antonio Castro (alcastro@isciii.es)
Oscar Moreno (omoreno@isciii.es)
Mario Pascual (mario.pascual@isciii.es)

Version: 1 Date: 02 Nov 2016

Author’s response to reviews:

In review of your "Availability of Data and Materials" section, you state "All data generated or analysed during this study are included in this published article.". Unfortunately, the data and associated source code do not appear to be included with your submission, and no link was provided.

I have corrected and updated the "Availability of Data and Materials" section to state the following: "Data sharing is not applicable to this article as no datasets were generated or analysed during the current study".

We have not analysed any datasets. We have just measured the response times to store, retrieve and query them in a DBMS.